brought to you by **CORE** 

ovided by Repositorio Institucional de la Biblioteca del Banco de E

2016

THE EFFECT OF ELECTORAL SYSTEMS ON VOTER TURNOUT: EVIDENCE FROM A NATURAL EXPERIMENT

Carlos Sanz

Documentos de Trabajo N.º 1623

# BANCODEESPAÑA

Eurosistema

THE EFFECT OF ELECTORAL SYSTEMS ON VOTER TURNOUT: EVIDENCE FROM A NATURAL EXPERIMENT

# THE EFFECT OF ELECTORAL SYSTEMS ON VOTER TURNOUT: EVIDENCE FROM A NATURAL EXPERIMENT

Carlos Sanz (\*)

PRINCETON UNIVERSITY

(\*) I thank Marco Battaglini and Thomas Fujiwara for their guidance and support. I also thank Jorge Álvarez, Jesús Fernández-Villaverde, Federico Huneeus, Matias laryczower, John Londregan, Fernanda Márquez-Padilla, José Miguel Rodríguez and seminar participants in Princeton University for their thoughtful comments and suggestions. I am solely responsible for any remaining errors.

Documentos de Trabajo. N.º 1623 2016

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2016

ISSN: 1579-8666 (on line)

#### Abstract

I exploit the unique institutional framework of Spanish local elections, where municipalities follow different electoral systems depending on their population size, as mandated by a national law. Using a regression discontinuity design, I compare turnout under closed-list proportional representation and under an open-list, plurality-at-large system where voters can vote for individual candidates from the same or different party lists. I find that the open-list system increases turnout by between one and two percentage points. The results suggest that open-list systems, which introduce competition both across and within parties, are conducive to greater voter turnout.

Keywords: voter turnout, electoral system, open list, regression discontinuity.

JEL classification: D72.

#### Resumen

En este artículo se explota el marco institucional de las elecciones municipales españolas, en las que se utilizan distintos sistemas electorales dependiendo del número de habitantes del municipio, según lo establecido por la Ley. Mediante una regresión discontinua, se compara la participación en un sistema de representación proporcional y listas cerradas con otro de listas abiertas, en el que los votantes pueden votar por candidatos individuales de la misma o diferentes listas. Se encuentra que el sistema de listas abiertas incrementa la participación electoral entre uno y dos puntos porcentuales. Los resultados sugieren que los sistemas de listas abiertas, que introducen competencia tanto entre como dentro de los partidos, conducen a más participación electoral.

Palabras clave: participación electoral, sistema electoral, listas abiertas, regresión discontinua.

Códigos JEL: D72.

# 1 Introduction

Voter turnout varies widely across countries. How much of that variation can be explained by differences in the electoral system? What is the causal effect of the electoral system on voter turnout? Do citizens participate more when they are allowed to decide which individual candidates get elected (open list systems), than in elections where they vote for a party-list (closed list systems)? These questions have attracted the attention of economists and political scientists for a long time (see Blais, 2006; Blais and Aarts, 2006; and Geys, 2006 for reviews of the literature).

There are two main characteristics of electoral systems: the degree of proportionality, determined mainly by the electoral formula and the district magnitude, and the ballot structure (open vs.closed lists). While the literature has extensively studied the former and generally agreed that more proportionality increases voter turnout (see, for example, Blais and Dobrzynska, 1998; Bowler et al., 2003; Fornos et al., 2004; Jackman and Miller, 1995; Ladner and Milner, 1999; Milner and Ladner, 2006; Schram and Sonnemans, 1996; Selb, 2009; St-Vincent, 2013), little is known about the effects of the ballot structure.<sup>1</sup> This is unfortunate, as changes between closed and open lists have been debated in several countries.<sup>2</sup> Understanding their influence on voter turnout is crucial to enlighten that debate and to the design of electoral systems.

Previous empirical evidence on this issue is mostly based on cross-country regressions, where small sample sizes and the endogeneity of electoral rules raise concerns about the causal interpretation of the estimates, as it is difficult to isolate the effect of the voting systems from other economic, cultural or institutional variables that may also affect voter turnout.<sup>3</sup> To the best of my knowledge, this is the first paper to use a quasi-experimental design to estimate how open list systems affect voter turnout. I focus on a previously unexplored setting in which electoral systems are exogenously assigned. In Spain, municipalities with more than 250 inhabitants elect a city council by closed list proportional representation while municipalities with 250 or fewer inhabitants elect a city council in a plurality-at-large, open list election, in which voters can vote for up to four individual candidates from the same or different party-lists. While the two systems differ in both proportionality and ballot structure (open vs. closed lists), in Section 5 I show that the difference in proportionality is small and provide evidence that the effect on turnout is driven by the difference in the ballot structure.

The institutional framework of Spanish local elections is a unique opportunity to study the causal effect of the electoral system on voter turnout for several reasons. First, the electoral system a municipality has to follow is determined by a *national* law as a function of the population size of the municipality, which reduces endogeneity concerns. Second, the number of observations is very high (around 72,000), as there are more than 8,000 municipalities in Spain and election results by municipality are available for 9 election years. Furthermore, there are many municipalities with a population size close to the population threshold that separates the two electoral systems (around 700 municipalities in a window of 50 inhabitants around the threshold). Third, all the municipalities under any of the electoral systems follow the exact same electoral system a set of systems that are only somewhat similar. Fourth, unlike what is often the case in situations where a policy changes at a municipal population threshold (Campa, 2012; Eggers, 2013; Grembi et al., 2012), no other rule changes at the threshold. Therefore, we can be confident in attributing the differences in outcomes between municipalities at each side of the threshold to the electoral system and not to some other regulation.

To carry out the analysis, I have collected a rich dataset with results from all elections held in Spain since the restoration of democracy in 1977. Combining regression discontinuity design and

 $<sup>^{1}</sup>$ As Blais and Aarts (2006) put it, "one aspect of electoral systems that has been neglected is the ballot structure" and "we know preciously little about the impact of ballot structure on turnout".

 $<sup>^{2}</sup>$ For example, Japan introduced an open list preference vote in elections for the upper house in 2001, while Italy abandoned an open list proportional representation system in 1993.

<sup>&</sup>lt;sup>3</sup>Countries that use open list systems include Belgium, Brazil, Chile, Colombia, Denmark, Finland, Indonesia, Iraq, Ireland, Italy, Japan, the Netherlands, Slovakia, Sweden and Switzerland.

fixed effects estimation, I find that the open list system increases voter turnout with respect to the closed list system by between one and two percentage points. There are many channels that may be conducive to these results - e.g. rational-choice calculations about the pivotality of votes or perceived fairness of the systems. I provide evidence that the differences in turnout are at least partially driven by the number of parties that enter competition. A higher number of parties in competition may in turn affect voter turnout by increased aggregate mobilization efforts and by providing voters with a more compelling set of options. I find that the open list system increases by 0.35 the average number of lists in competition. This effect is most likely driven by the fact that the open list system makes it much easier for popular candidates from small parties to get elected, as voters can choose individual candidates from the same or different party lists.

An issue that requires special attention is that there exists some evidence that municipalities may be able to partially control their population size, as some sorting is observed around the threshold. I carefully study this question in Section 6, assessing the validity of the empirical strategy and checking the robustness of the results to donut-regression discontinuity design estimation in the spirit of Barreca et al. (2011), by dropping observations within a window where the sorting is most likely to occur.

The rest of the paper is organized as follows. Section 2 reviews the literature on the impact of electoral systems on voter turnout, analyzing both theoretical predictions and the available empirical evidence. Section 3 provides background on the Spanish electoral systems. The data and empirical strategy are laid out in Section 4. Section 5 presents the main results and looks into the mechanisms at work. The robustness of the results is analyzed in Section 6. Section 7 concludes.

# 2 Literature Review

This paper contributes to the literature that tries to explain voter turnout (see, for example, Downs, 1957, and Riker and Ordeshook, 1968) and, in particular, to the literature that studies the link between the electoral system and voter turnout. The electoral systems that I compare in this paper differ mainly in their ballot structure. In this section, I review the theoretical arguments and the previous empirical evidence on the effect of the ballot structure on voter turnout.

In some elections, voters can express a preference for candidates within the party lists (open list systems, OL), while in others they are limited to choose between different lists (closed list systems, CL). There are opposing views in the literature about how the use of OL vs. CL systems should affect voter turnout.

On the one hand, it has been argued that OL systems should increase turnout. Mattila (2003) argues that since voters can choose the candidate they wish to vote for, they are likely to feel more satisfied with the act of voting. Along the same lines, Karvonen (2004) indicates that voting for individual candidates makes the election more personal and concrete and that both elements should provide a stimulus for active electoral participation. Supporting this hypothesis, Hix and Hagemann (2009) find that citizens in EU states who use OL are almost five percent more likely to be contacted by candidates or parties than citizens in member states with CL systems. They are also more than 20 percent more likely to be contacted by candidates or parties and about 15 percent more likely to feel well informed about the elections than citizens in states with CL systems.

On the other hand, Robbins (2010) hypothesizes that turnout should be higher in CL systems. His argument is that in OL systems parties may not exert the same level of resources to solicit support or mobilize voters as in CL systems. Individual candidates, for their part, will appeal to their supporters but will likely avoid mobilization strategies that involve the entire population. In CL systems, on the contrary, "parties place greater emphasis on mobilizing voters everywhere in hopes of soliciting additional support. After all, if they construct the list, then they are responsible for the success of their candidates and will devote more time, energy and resources calling individuals to the polls".

The empirical evidence for the effect of the ballot structure is scarce, probably due to limited cross-country variation. Hix and Hagemann (2009) find that voters are almost 10 percent more

likely to cast their votes on election day in OL systems. Robbins (2010), on the contrary, finds that OL decreases turnout levels. Mattila's (2003) empirical findings, with data from elections to the European Parliament, show that a variable indicating a CL system is not significant. The empirical analyses in Blais and Aarts (2006), Dos Santos (2007) and Karvonen (2004) also conclude that there is not sufficient evidence to support the hypothesis of a positive correlation between preferential voting and electoral participation. Eggers (2013) uses a discontinuity in the electoral rules of French local elections and finds that a closed list proportional representation system leads to more turnout than an open list plurality system, but he attributes the effect to the different proportionality of the systems. In sum, the empirical evidence is non-conclusive: the effect of the ballot structure on turnout is still an open question.

Finally, another strand of literature has studied the effects of OL systems on other outcomes.<sup>4</sup> Farrell and McAllister (2006) find that preferential voting systems where voters are given more freedom in completing the ballot paper lead to higher satisfaction with democracy. Other papers have found that OL systems increase the value of personal reputation with respect to party reputation by enhancing intra-party competition and electoral uncertainty (Carey and Shugart, 1995, Chang, 2005) or by inducing voters to focus more on candidates' characteristics and less on parties' positions (Shugart et al., 2005). Ames (1995a, 1995b) studies the use of OL elections in Brazil and supports these conclusions by highlighting the very weak role played by national parties in the country. Persson et al. (2003) find that OL systems reduce corruption, while Chang and Golden (2007) find the opposite effect. Negri (2014) develops a theoretical model and predicts that, in general, CL proportional representation is associated to lower minority representation within Parliaments than OL systems.

# 3 Spanish Electoral Systems

Spain is politically decentralized in 17 regions and more than 8,000 municipalities. Each municipality elects a local government in free elections. A national law requires local governments to provide a variety of services, including public lighting, waste collection, cemeteries, street cleaning and road pavement (see Table 1 for a comprehensive list). In addition, municipalities are allowed to provide any other service that they consider useful to the municipality. For example, it is common that they provide touristic information to visitors and organize local festivities. Municipalities can levy a number of taxes and charges (most importantly, a property tax) and receive transfers from regional and national governments to finance some of their expenditures. Approximately 55 percent of their revenues come from the taxes imposed by themselves (Sweeting, 2009).

Local elections are held simultaneously in all municipalities in Spain every fourth year.<sup>5</sup> The elections follow one of three electoral systems, depending upon the population size of the municipality.

Municipalities with more than 250 inhabitants use a closed list proportional representation system (henceforth, the CL system).<sup>6</sup> They elect a city council in a single-district, closed list election, where each party presents a list of candidates and citizens can vote for one of the party lists. The size of the council increases with population at certain population cutoffs but all municipalities in the CL system used for identification elect a 7-member council (as the empirical strategy relies on municipalities close to the threshold). The conversion from votes to seats is done according to the D'Hondt rule.<sup>7</sup> The council elects a mayor among its members and is entitled to approve the budget, decide on expenditure in various fields, control the governing bodies and to the roll-call

<sup>&</sup>lt;sup>4</sup>This discussion is partially based on the literature review in Negri (2014).

<sup>&</sup>lt;sup>5</sup>Local elections coincide with regional elections in 13 of the 17 Spanish regions and in one year (1999) they also coincided with elections for the European Parliament. The implications of this are studied in the empirical strategy. <sup>6</sup>The population size that determines the electoral system is the official population in the municipal register (*padrón municipal*) the 1st of January of the year before the elections.

 $<sup>^{7}</sup>$ There is an electoral threshold at 5 percent, i.e. parties need to get at least 5 percent of the votes to enter the D'Hondt distribution of seats. However, given the size of the council of the municipalities we are studying, that threshold does not play an important role in these elections.

Municipalities	Services
All	Public lighting, cemeteries, waste collection, street
	cleaning household water supply, sewerage, access to
	villages, paving roads, food and beverage control
> 5000	Public park, public library, waste treatment,
	organization of markets
> 10000	Civil protection, social services, prevention and
	fire-fighting, sports facilities for public use
> 20000	Urban public passenger transport and environmental
	protection

Table 1: Services of Spanish municipalities by population

vote of confidence on the mayor. The mayor chairs the meetings of the council, casts the decisive vote in the event of a tie, decides on some expenditures, heads the local police and appoints mayoral deputies and cabinet members, among other responsibilities.

Municipalities with a population between 100 and 250 inhabitants follow an open list, pluralityat-large system (henceforth, the OL system). Under this system, a council of five members is elected. Political parties can present candidate-lists of up to five candidates and voters can vote for up to four candidates from the same or different party lists. The five most voted candidates are elected members of the council.<sup>8</sup> As in the CL system, the council elects a mayor among its members and the responsibilities of the council and mayor are identical under the two systems.

Finally, municipalities with fewer than 100 inhabitants directly elect a mayor in a simple plurality, first-past-the-post election, i.e. each political party can present one candidate and the most voted candidate is elected mayor. These municipalities follow a direct democracy system in which the role of the council is played by open meetings that any citizen in the municipality can attend.<sup>9</sup>

The paper focuses on the 250-inhabitant threshold and compares voter turnout under the CL and OL systems, as only the electoral system changes at that threshold. The results for the 100-inhabitant threshold are discussed in Section 5.

The CL and OL systems differ in two main dimensions. First, the electoral formula differs as under the OL system seats are allocated by plurality to the most voted individual candidates, while under CL they are allocated to parties according to the D'Hondt rule. Both the change in the electoral formula and the increase in the council size (from 5 to 7 members) imply that the CL is a more proportional system *at the party level*. Second, the two systems differ on the ballot structure: while the CL system is a closed list system where competition is limited to across-parties competition, the OL system allows voters to express their preferences for individual candidates, making it easier for popular candidates from small parties to get a seat in the council and introducing competition both across and within parties. In Section 5, I provide evidence that the difference in proportionality is small in practice, and that it is the difference in ballot structure what drives the results, with open lists increasing voter turnout.

Note: The table shows the services that Spanish municipalities are required to provide, as a function of population size. Source: Law 5/1985 and Campa (2012).

 $<sup>^{8}</sup>$ The fact that candidates can vote for fewer candidates than there are seats to elect means that it is a *limited voting* system.

 $<sup>^{9}</sup>$ Municipalities with more than 100 inhabitants could decide to adopt this system. In order to do that, a majority of the citizens of the municipality had to sign a petition and two thirds of the members of the council and the regional government had to approve it. To the best of my knowledge, no municipality ever used this procedure. Therefore, the regression discontinuity design can be sharp, as the probability of treatment jumps from 0 to 1 at the threshold.

# 4 Empirical Strategy and Data

#### 4.1 Empirical Strategy

To estimate the effect of the electoral system on voter turnout, I combine regression discontinuity and fixed-effects estimation. In particular, I consider the following estimating equation:

$$y_{mt} = \alpha_m + \gamma_t + \beta D_{mt} + f(x_{mt} - x^*) + u_{mt},\tag{1}$$

where  $y_{mt}$  is the outcome of interest (in the main specifications, voter turnout),  $D_{mt}$  is a treatment dummy that captures the electoral system municipality m is required to follow in election year t,  $x_{mt}$  is the assignment variable (log population the year before the elections)<sup>10</sup>,  $x^*$  is log of the population threshold (250 inhabitants), so that treatment status depends on whether  $x_{mt}$  is bigger or smaller than  $x^*$ , f is a smooth function of the assignment variable,  $\alpha_m$  is a municipality fixed effect,  $\gamma_t$  is a year fixed effect and  $u_{mt}$  is an error term. The parameter of interest is  $\beta$ .<sup>11</sup>

To estimate f, I use non-parametric estimation (local linear regression). A key ingredient to this approach is the bandwidth. There is a trade-off between precision and bias: a bigger bandwidth increases precision at the cost of more bias. I choose a baseline bandwidth according to the procedure suggested by Imbens and Kalyanaraman (2012) and provide the results at different fractions of that bandwidth to see how sensitive the results are to bandwidth choice. Notice that we may not want the bandwidth to be bigger than 150 inhabitants (remember that municipalities with fewer than 100 inhabitants follow a different system), as if that happened we would be mixing outcomes from the three electoral systems in the same specification. For that reason, I restrict the analysis to bandwidths of less than 150 inhabitants. I use a rectangular kernel, as recommended by Imbens and Lemieux (2008) and Lee and Lemieux (2010). This is equivalent to estimating standard linear regressions over the interval of the selected bandwidth on both sides of the cutoff point. I cluster the standard errors at the municipality level.

A possible cause of concern in a pure regression discontinuity framework would be that a discontinuity in the density of population size is observed at the threshold (Lee and Lemieux, 2010) (see Figure 1).<sup>12</sup> To deal with that issue, I add municipality and year fixed effects to the basic regression discontinuity framework. The identification assumption is that there are no unobservable factors that may simultaneously affect voter turnout and whether a municipality's population is just above or just below the threshold, *conditional on municipality and year fixed effects*. The identification therefore relies on switchers: intuitively, the regressions do not compare municipalities just above and just below the threshold but municipalities that switch from one system to another with those that remain in the same system.<sup>13</sup> If the factors that make municipalities sort around the threshold are time invariant,  $\beta$  will identify the average treatment effect of the electoral system on voter turnout for municipalities close to the threshold.<sup>14</sup>

Although the identification assumption is fundamentally untestable, in Section 6 I present three sets of tests to address the validity of the strategy. First, I test whether municipalities at each side

 $<sup>^{10}\</sup>mathrm{Geys}$  (2006) recommends using log population for turnout studies.

<sup>&</sup>lt;sup>11</sup>Similar strategies to the one used in this paper have been used by Almunia and Lopez-Rodriguez (2012), Grembi et al (2012), Lemieux and Milligan (2008) and Pettersson-Lidbom (2012). More generally, population thresholds have been widely used in recent years as a way to get a credible estimation of causal effects (Arnold and Freier, 2013; Bordignon et al, 2013; Brollo et al, 2009; Campa, 2012; Casas-Arce and Saiz, 2012; Egger and Koethenbuerger, 2010; Eggers, 2013; Ferraz and Finan, 2009; Fujiwara, 2011; Fujiwara, 2015; Gagliarducci and Nannicini, 2013; Hinnerich and Pettersson-Lidbom, 2014; Hirota and Yunoue, 2013; Litschig and Morrison, 2013; Petterson-Lidbom, 2006).

 $<sup>^{12}</sup>$ The discontinuity is significant using McCrary's (2008) test in a pooled cross section of all the municipality-years, but not if each election-year is considered separately. In Section 7 I describe in detail why and how that discontinuity appears.

 $<sup>^{13}</sup>$ The inclusion of fixed effects implies that municipalities just above and just below the threshold *can* differ in other factors that affect voter turnout.

<sup>&</sup>lt;sup>14</sup>The use of fixed effects implies that it is not possible to provide a graphical representation of the results, as the comparison of municipalities above and below the threshold is conditioned on the municipality and year fixed effects. (See Pettersson-Lidbom (2012) for a paper that combines regression discontinuity and fixed effects and can't provide a graphical analysis.)



of the threshold differ, conditional on the fixed effects, in other variables that may themselves affect the outcomes of interest. Second, I estimate a dynamic model to test for pretrends: in particular, I study whether this period's electoral system has an effect on previous period's turnout. Third, I consider donut regressions to examine the robustness of the results to the exclusion of some observations where any problem of self-selection that might remain after the inclusion of the fixed effects is likely to be concentrated.

Another issue that requires some consideration is that local elections are held on the same day as regional elections in 13 of the 17 Spanish regions. In 1999, local elections also coincided with elections to the European Parliament. This can be analyzed as a measurement error issue. Observed turnout  $t_{mt}$ , is measured with error as it differs from the "true" turnout  $t_{mt}^*$ , which is the turnout that would have been observed if local elections had been the only elections on election day. Let the error be  $e_{mt} = t_{mt} - t_{mt}^*$ . If (1) captures the true relationship between the outcome and independent variables, then  $y_{mt} = t_{mt}^*$ .

Thus, the estimated model is

$$t_{mt} = \alpha_m + \gamma_t + \beta D_{mt} + f(x_{mt} - x^*) + u_{mt} + e_{mt}.$$
 (2)

If the new error term  $u_{mt} + e_{mt}$  is uncorrelated with the regressors, the estimators will be unbiased and consistent. Thus, if  $e_{mt}$  is not correlated with the electoral system (and there is no reason to think it is, because the population threshold does not play any role in regional or European elections), measurement error will not affect the unbiasedness and consistency of the estimators.<sup>15</sup>

#### 4.2 Data

I have collected data from all local elections that have taken place in Spain since the restoration of democracy after General Franco's death in 1975. In this period, local elections have been held in 9 years (1979, 1983, 1987, 1991, 1995, 1999, 2003, 2007, 2011) in all municipalities in the country (around 8000).<sup>16</sup> In addition to the data from local elections, I have collected data at the

 $<sup>^{15}</sup>$ The estimators will nonetheless be inefficient, but the big sample size helps to overcome that problem.

<sup>&</sup>lt;sup>16</sup>Due to some missing values or errors in the official elections data, the final dataset used in this paper contains observations for 71,780 out of the approximately 73,026 local elections in this period, i.e. 98.6 percent of the elections. I cannot know the number of total elections because for the first years in the sample I only observe municipalities with elections data, so I do not know how many municipalities held elections that are not in my sample. I calculated 73,026 by assuming that the number of municipalities remained constant over time.

municipality level from all national Congress elections in that period.<sup>17</sup> These data will be used in Section 6 to analyze the validity of the identification strategy. All data are from the National Statistics Institute (INE) and are publicly available.

Table 2 presents summary statistics for the variables used in the paper. The main outcome of interest is *Turnout*, defined as the number of votes cast divided by the electoral census. That is, *Turnout* measures the proportion of citizens that cast a vote over the set of potential voters (the electoral census). There is no voter registration in Spain: potential voters are all citizens from Spain, EU countries and countries under Reciprocity Treaties, older than 18 and not disenfranchised by court order.<sup>18</sup> The other outcome variable of interest is *Lists*, defined as the number of party-lists that run for election.

I consider six variables from Congress elections: the percent of voter turnout (N turnout), defined analogously to the one for local elections, the percent of blank  $(N \ blank)$  and spoilt  $(N \ spoilt)$  votes, defined as the number of blank and spoilt votes divided by the number of votes cast, and the share of votes for the three main parties in Spain: the right-wing Popular Party  $(N \ right)$ , the left-wing Socialist Party  $(N \ left)$  and the far-left United Left  $(N \ far \ left)^{19}$ , defined as the votes for these parties divided by the number of valid votes.<sup>20</sup>

### 5 Results

#### 5.1 Main Results

Table 3 presents the main estimates of the impact of the electoral system on voter turnout. The table shows the results of estimating equation (1) by local linear regression, using *Turnout* as the dependent variable. The treatment variable is OL, which takes the value of 1 if the municipality follows the OL system and 0 if it follows the CL system. The coefficient on OL therefore captures the effect on turnout of using the OL system in relation to the CL system, i.e. it measures the effect on turnout of crossing the threshold from right to left.

Column (1) provides the estimated effect based on the preferred specification, which uses the bandwidth suggested by Imbens and Lemieux (2008). Under that specification, the OL system increases turnout in relation to the CL system by one percentage point, and the effect is statistically significant at the 1 percent level. The rest of the columns show the results for different fractions of that bandwidth. The effect is statistically significant and quantitatively similar across bandwidths, ranging from one to two percentage points, suggesting that the effect is not driven by the choice of the bandwidth. In conclusion, the results imply that the OL system leads to more voter turnout than the CL system.

#### 5.2 Additional Results: Mechanism

The OL and CL systems differ in two main dimensions. First, the electoral formula differs as under the OL system seats are allocated by plurality to the most voted individual candidates, while under the CL system they are allocated to parties according to the D'Hondt rule. Both the change in the electoral formula and the increase in the council size (from 5 to 7 members) imply that the CL is a more proportional system *at the party level*. Second, the two systems differ on the ballot structure:

 $<sup>^{17}</sup>$ Spain is a bicameral parliamentary system. The national elections data are for elections to the Lower House (*Congreso*), which concentrates most of the political power. Throughout the paper, "national elections" refer to these elections.

 $<sup>^{18}</sup>$  Disenfrachisement is mostly for disability reasons. In 2011 the figure of disenfranchised individuals was 79398 (including individuals younger than 18), or around 0.18 percent of the population.

 $<sup>^{19}</sup>$ United Left (*Izquierda Unida*) is a coalition of parties created in 1986 whose main party is the Communist Party. For elections before that date, votes for the Communist Party are considered. The sample size is smaller than for the main parties because, in some years, the coalition did not run in some regions.

 $<sup>^{20}</sup>$ Valid votes include votes for candidates and blank votes, but not spoilt votes. I use this denominator because it is the relevant one for the allocation of seats, as it is used to determine whether parties reach the election threshold to get seats (3 percent in Congress elections). Accordingly, it is the one that is normally reported by the media.

 Table 2: Summary Statistics

	mean	$\operatorname{sd}$	$\min$	p10	p50	p90	max	count
All municipalities								
turnout (%)	75.4	12.1	0.90	59.6	77.4	88.7	100	71780
lists $(\%)$	2.90	1.56	1.00	1	3	5	25	71780
n turnout (%)	77.2	8.97	0.80	66.0	78.4	87.0	100	63478
n blank (%)	0.88	1.23	0.00	0	0.59	1.99	50	63478
n spoilt (%)	1.07	1.93	0.00	0	0.62	2.45	84.6	63478
n right (%)	39.5	19.5	0.00	12.4	40.1	64.5	100	63478
n left (%)	37.3	15.8	0.00	15.9	37.8	57.5	100	63478
n far left $(\%)$	3.73	4.51	0.00	0	2.43	8.54	66.7	56233
OL municipalities	$(100 \le p$	oopulat	ion $\leq 2$	50).				
turnout (%)	77.4	11.9	0.90	61.5	79.7	90.1	100	13496
lists (%)	2.19	0.91	1.00	1	2	3	6	13496
n turnout (%)	77.9	8.65	2.11	67.1	79.0	87.6	100	11941
n blank (%)	1.02	1.46	0.00	0	0.71	2.68	50	11941
n spoilt $(\%)$	1.15	2.18	0.00	0	0	3.23	45.0	11941
n right (%)	45.8	20.5	0.00	14.6	48.1	71.3	97.5	11941
n left (%)	31.2	14.8	0.00	11.9	30.8	50.5	87.9	11941
n far left $(\%)$	2.68	3.16	0.00	0	1.72	6.52	43.8	10515
CL municipalities	$(251 \le p)$	opulat	$ion \leq 40$	00).				
turnout (%)	76.6	14.0	6.54	56.2	80.4	90.3	100	8059
lists (%)	2.08	0.77	1.00	1	2	3	6	8059
n turnout (%)	77.5	8.85	13.31	66.1	78.9	87.2	99.3	7009
n blank (%)	0.89	0.96	0.00	0	0.61	2.11	11.0	7009
n spoilt (%)	1.11	1.94	0.00	0	0.55	2.68	52.4	7009
n right (%)	40.7	19.5	0.00	11.3	43.1	64.6	91.4	7009
n left (%)	34.6	15.1	0.00	14.0	35.4	53.4	81.0	7009
n far left $(\%)$	2.68	2.98	0.00	0	1.88	6.06	50	6237

The first panel shows summary statistics for all municipalities.

The second panel shows summary statistics for municipalities under the OL system. The third panel, for those under the CL system and fewer than 400 inhabitants,

which is the biggest bandwidth in the regressions.

Table 3: Effect of Ballot Structure on Voter Turnout: Open vs. Closed List

	(1) turnout	(2) turnout	(3) turnout	(4) turnout	(5) turnout	(6) turnout
						- t accelu
OL	$1.031^{***}$ (0.381)	$1.237^{***}$ (0.395)	$1.156^{***}$ (0.407)	$1.547^{***}$ (0.431)	$1.844^{***}$ (0.491)	$1.308^{**}$ (0.641)
Observations	15954	13556	11226	8798	6404	3976
Municipalities	2826	2597	2343	2079	1803	1512
Bandwidth	IK	.85*IK	.70*IK	.55*IK	.40*IK	$.25^{*}IK$

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants.

\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

while the CL system is a closed list system where competition is limited to competition across political parties, the OL system allows voters to express their preferences for individual candidates, introducing competition both across and within parties. In this section, I provide evidence that the ballot structure difference is more relevant and the one that drives the results.

First, although the CL system is formally a proportional representation system, in practice it is only slightly more proportional than the OL system. Under the OL system, only in 3.5 percent of the elections doesn't the most voted party get an absolute majority (3/5 or more) of the seats (and therefore most of the political power, including the possibility of appointing the mayor).<sup>21</sup> But the cases in which no party has an absolute majority of seats in the council are also rare under the CL system, around 7 percent of the elections. That is because the number of seats is small (7) and the D'Hondt rule is used, which favors the most voted parties when the district size is small.

Second, I study whether the electoral system also has an effect on the number of party-lists running in the election. Panel A of Table 4 shows that the OL system increases the number of lists by around 0.35, which suggests that at least part of the change in voter turnout might be driven by entry decisions of candidates. A higher number of parties in competition may in turn affect voter turnout by increased aggregate mobilization efforts and by providing voters with a more compelling set of options.<sup>22</sup>

Panel B of Table 4 shows the results for voter turnout after a control for the number of lists is included in the main regression. The coefficient on *Lists* indicates that one more party running in the election increases turnout by around six percentage points. But the more interesting result is that the estimated effect of the electoral system reverses when the control for the number of lists is

 $<sup>^{21}</sup>$ To see why that is the case, consider a scenario in which there are three parties, A, B and C, and that 40% of the electorate prefers party A, 35% party B and 25% party C. Assume that voters are indifferent between the two parties other than their preferred party and, to focus on across-parties competition, that they do not care about individual candidates but only about their preferred party getting the mayor. In this case, party A can present a list of four candidates and its supporters give their four votes to those four candidates. This guarantees 4 seats to party A, no matter how many candidates parties B and C decide to include on their lists. Party B will get the other seat unless it decides to include 5 candidates on its list. Thus, even with a minor advantage in terms of votes, the leading party gets 4 out of 5 seats and therefore the mayor. If the advantage is sufficiently big, it may get the 5 seats by presenting 5 candidates if voters split their votes sufficiently across candidates.

 $<sup>^{22}</sup>$ It has also been argued that a higher number of parties could decrease voter turnout as it usually produces coalition governments, which make elections less decisive (Jackman, 1987). However, that is unlikely to play a big role in the case studied in this paper because coalition governments are rare in the two systems that I compare, as is explained above.

introduced: conditional on the number of lists, turnout is higher under the CL system. Of course, this does not mean that the causal effect is actually positive. The number of lists is an endogenous or "bad" control in a regression whose goal is to find the true causal effect, as it is also an outcome of the electoral system (see Angrist and Pischke (2008) on bad controls). However, this exercise reinforces the argument that the reason why the OL system leads to more turnout than the CL system is that it encourages more parties to enter competition.

It is hard to explain the difference in number of parties as being driven by the different proportionality of the systems: if voters care only about party-level competition, the CL system makes it easier for small parties to get some representation so we should expect more parties to run under the CL system.<sup>23</sup> The change in the number of parties is most likely driven by the different ballot structure of the systems, as the OL system makes it easier for individual candidates in small parties to get elected to the council. For example, assume there are three main parties A, B and C, and a small party D that has a very popular leader, and that voters care both about parties and individual candidates. Suppose that if voters can only vote for one party-list, as in the CL system, most individuals prefer to vote for parties A, B and C, so these three parties share the 7 seats. That is, although D has a very popular leader, most voters prefer to vote for their preferred parties A, B and C if they can only vote for one party-list. In this scenario, party D will not enter competition if there is a cost of running. On the other hand, under the OL system, voters are allowed to vote for four individual candidates, from the same or different party-lists. Thus, voters could give three of their votes for candidates of their preferred party and one vote for the leader of party D. Under this system, party D will enter competition if the benefit of getting a seat is bigger than the cost of running. This argument suggests that the ballot structure of the OL system, which allows voters to choose among candidates from different party-lists, should lead to more parties in competition.

Third, I compare the OL system with a third electoral system, exploiting a discontinuity in electoral systems at a different threshold: municipalities with 100 or more inhabitants (but with not more than 250) follow the OL system, while those with fewer than 100 inhabitants elect a mayor in a first-past-the-post (FPTP) election, i.e. each political party can present one candidate and the most voted candidate is elected mayor. These municipalities follow a direct democracy system in which the role of the council is played by open meetings that any citizen in the municipality can attend.<sup>24</sup> Although the estimates for this threshold should be interpreted with caution, as in the 100-inhabitant threshold there is a change in the government system in addition to the change in the electoral system, the evidence presented here reinforces some of the conclusions drawn from the main results from the 250-inhabitant threshold. In particular, if it is the ballot structure and not the proportionality of the systems what drives the increased turnout under the OL with respect to the CL system, we would also expect that effect to be present when we compare the OL system with FPTP, as these two systems also differ on the ballot structure but are both plurality (and not proportional representation) systems.

To study turnout at this threshold, I follow the same empirical strategy described in the previous section. The treatment variable is (slightly abusing notation) again OL, which takes the value of 1 if the municipality follows the OL system and 0 if it follows the FPTP system. The results in Panel A of Table 5 show that the OL system increases voter turnout by between two and three percentage points with respect to FPTP.<sup>25</sup> It also increases the number of lists in competition, although by a

 $<sup>^{23}</sup>$ Indeed, it has been generally accepted that proportional representation systems increase the number of parties in competition (Blais and Aarts, 2006).

 $<sup>^{24}</sup>$ Sanz (2014) uses the change in the government system at the 100-inhabitant threshold to study how direct democracy, as opposed to representative democracy, affects public finances.

 $<sup>^{25}</sup>$ For this threshold I exclude the observations from 1979, 1983 and 2011 because the threshold was not relevant to the electoral system in those years. In 1979 and 1983, the threshold for electing a mayor by FPTP was at 25 inhabitants rather than at 100, so that system was virtually not used (there was only one municipality with a population of fewer than 25 inhabitants in those two years). A change in the electoral law in 2011 determined that all municipalities had to elect a council in the 2011 election so no FTPT elections were held that year. I also exclude the observations from 1987. These were the first elections in which all the municipalities with fewer than 100 inhabitants were required to follow the FPTP system and in the majority of municipalities only one list ran for the election, probably as a consequence of parties not having time to adapt to the new system. To prevent these transitional effects from affecting the results I do not include observations from 1987 in the estimation. In any case, the results

Table 4: Effect of Ballot Structure on Number of Party-Lists and on Voter Turnout When Controlling for Party-Lists

Panel A: Party-Lists

	(1)	(2)	(3)	(4)	(5)	(6)
	lists	lists	lists	lists	lists	lists
OL	$0.314^{***}$	$0.317^{***}$	$0.331^{***}$	$0.355^{***}$	$0.387^{***}$	0.382***
	(0.0292)	(0.0304)	(0.0321)	(0.0341)	(0.0402)	(0.0511)
Observations	17693	15148	12517	9845	7114	4472
Municipalities	2993	2744	2483	2196	1885	1577

Panel B: Turnout When Controlling For Party-Lists

	(1)	(2)	(3)	(4)	(5)	(6)
	turnout	turnout	turnout	turnout	turnout	turnout
lists	$6.042^{***}$	$5.979^{***}$	$5.916^{***}$	$6.063^{***}$	$5.937^{***}$	$5.956^{***}$
	(0.152)	(0.169)	(0.189)	(0.216)	(0.256)	(0.319)
OL	-0.860**	-0.708**	-0.857**	-0.655	-0.578	-0.944
	(0.343)	(0.359)	(0.373)	(0.403)	(0.453)	(0.597)
Observations	15954	13556	11226	8798	6404	3976
Municipalities	2826	2597	2343	2079	1803	1512
Bandwidth	IK	.85*IK	.70*IK	.55*IK	.40*IK	.25*IK

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants. \* p<0.10, \*\* p<0.05, \*\*\* p<0.01.

Table 5: Effect of Ballot Structure on Voter Turnout and Party-Lists: Open List vs. FPTP (100-inhabitant threshold)

Panel A: Turnout

	(1)	(2)	(3)	(4)	(5)	(6)
	turnout	turnout	turnout	turnout	turnout	turnout
OL	$2.811^{***}$	$2.808^{***}$	$2.566^{***}$	$2.447^{***}$	$2.474^{***}$	2.243***
	(0.521)	(0.531)	(0.548)	(0.580)	(0.656)	(0.807)
Observations	11161	9662	8020	6297	4604	2901
Municipalities	2621	2349	2054	1714	1384	1057
Bandwidth	IK	$.85^{*}IK$	$.70^{*}IK$	.55*IK	$.40^{*}IK$	.25*IK

Panel B: Party-Lists

	(1)	(2)	(3)	(4)	(5)	(6)
	lists	lists	lists	lists	lists	lists
OL	$0.0761^{*}$	$0.0764^{*}$	$0.0808^{*}$	0.0462	0.0764	$0.143^{*}$
	(0.0390)	(0.0408)	(0.0433)	(0.0491)	(0.0533)	(0.0788)
Observations	6957	5952	4949	3924	2901	1874
Municipalities	1833	1649	1462	1253	1057	835
Bandwidth	IK	.85*IK	$.70^{*}IK$	.55*IK	.40*IK	.25*IK

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants. \* p<0.10, \*\* p<0.05, \*\*\* p<0.01.

modest amount (0.1) (Panel B). Therefore, the results from this threshold go in the same direction as those obtained from the main threshold: not only does the OL system lead to more turnout and more lists in competition than the CL system, but the OL system also produces the same effects when we compare it with FPTP.<sup>26</sup>

# 6 Robustness

As explained in Section 4, a discontinuity in the density of population sizes is observed at the threshold. In the first part of this section, I review how that discontinuity arises. In the subsequent parts of the section present three sets of robustness checks to assess the validity of the empirical strategy.

# 6.1 Sorting around the Threshold: Process

The official population size of a municipality is given by the number of citizens who are registered in the municipal register (*padron municipal*). Municipalities keep track of all the variations in the population (births, deaths, registrations and unregistrations) in the public register and report periodically the data to the National Statistics Institute (INE). The INE validates the information it receives, checking that there is no fraud –for example, it makes sure that for every sign-in in a municipality there is a corresponding sign-out in another- and yearly makes the final population figures public. Given the mechanics of the process, it is hard to believe that there is outright manipulation of the population figures. Instead, the discontinuity at the threshold arises because individuals that have dwellings in more than one municipality can in practice decide in which of them to register.<sup>27</sup> In particular, the mayor or other local politicians may be interested in the electoral system that the municipality follows and may make an effort to persuade some individuals to register in the municipality so that it keeps the preferred electoral system.<sup>28</sup> Therefore, some individuals that would have registered in another municipality if there was no change of rules at the threshold, decide to register in the municipality.

Although the identification assumption is fundamentally untestable, I now present three sets of tests to assess the validity of the strategy. First, I test whether municipalities at each side of the threshold differ, conditional on the fixed effects, in other variables that may themselves affect the outcomes of interest. Second, I estimate a dynamic model to test for pretrends: in particular, I study whether this period's electoral system has an effect on previous period's turnout. Third, I consider donut regressions to examine the robustness of the results to the exclusion of some observations where any problem of self-selection that might remain after the inclusion of the fixed effects is likely to be concentrated.

# 6.2 Placebo Tests: Covariates at the Thresholds

The identification strategy requires municipalities that municipalities that switch into the OL system be comparable with those that stay in the same system and with those that switch out of the OL system. In other words, municipalities just above and just below the threshold should not differ, conditional on the fixed effects, in other variables that may themselves have an effect on voter turnout. Here I use national Congress election variables and test whether they change at

are still significant (and in general even bigger in magnitude) if that year is included in the estimation.

 $<sup>^{26}</sup>$ The robustness checks for the results from this threshold are presented in the Appendix.

<sup>&</sup>lt;sup>27</sup>Although individuals who live in more than one municipality are required to register in the municipality in which they spend more time, that requirement is almost impossible to monitor and not enforced in practice.

 $<sup>^{28}</sup>$ In particular, the spike of the distribution of population sizes suggests that there is a preference to fall under the CL system. A possible reason for this preference is that the number of seats increases at the threshold, therefore leading to more political positions available. Indeed, similar spikes on the distribution are observed at other thresholds where the council size increases, although it is not possible to know whether the spike is entirely caused by the council size as more regulations change at the other thresholds.

the thresholds by including them as dependent variables in equation (1).<sup>29</sup> These covariates are especially suited to this context because if municipalities differ in some unobservable factors that affect voter turnout at local elections, it is likely that those unobservable variable affect turnout at national elections too. Furthermore, Congress elections are the most important elections in Spain and turnout is typically very high (78 percent in the average municipality during the sample period) so they are likely to capture any political differences across municipalities.<sup>30</sup>

As pointed out by Lee and Lemieux (2010), it is important that the covariates are determined prior to the present period's realization of the running and treatment variables if they can be affected by treatment.<sup>31</sup> I merge local elections data with data from the most recent previous Congress elections at the municipality level.<sup>32</sup> I consider the percent of voter turnout in (*N turnout*), blank (*N blank*) and spoilt votes (*N spoilt*) in Congress elections and the percent of votes for the three main parties in Spain (*N right*, *N left* and *N far left*).<sup>33</sup> If the empirical strategy is valid, there should be no effect of the electoral system on these variables.

The results from these tests are shown in Table 6. Among the six variables considered, only in one (the share of spoilt votes) there seems to be some difference between municipalities, but the effect is quantitatively small and only significant in some specifications. Overall, the results suggest that, conditional on the fixed effects, municipalities just above and just below the threshold do not differ sistematically along these variables: while municipalities operating under open and closed lists behave differently in local elections, they do not do so in national elections.

 $<sup>^{29}</sup>$ Note that the inclusion of the fixed effects implies that to test the robustness of the strategy we need variables that change over time.

 $<sup>^{30}</sup>$ Spain follows a parliamentary system, so there are no elections for the executive branch. Citizens elect the Congress, which in turn elects the Prime Minister.

 $<sup>^{31}</sup>$ Although the population threshold does not play any role in national elections, behavior at subsequent national elections could be affected if, for example, the electoral system influences voters' political views.

 $<sup>^{32}</sup>$ Thus, I estimate equation (1) where instead of turnout at the 1983 (1987, 1991, 1995, 1999, 2003, 2007, 2011) local elections I include turnout at the 1982 (1986, 1989, 1993, 1996, 2000, 2004, 2008) national elections as the dependent variable.

 $<sup>^{33}</sup>$ The other parties represented in Congress during the whole period are regional parties that do not run in the whole country.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	n turnout	n turnout	n turnout	n turnout	n turnout	n turnout	n blank					
OL	-0.0562	-0.185	-0.212	-0.172	-0.0113	0.375	0.00401	0.00638	0.0306	0.0259	0.0651	0.0531
	(0.192)	(0.193)	(0.199)	(0.210)	(0.222)	(0.250)	(0.0337)	(0.0347)	(0.0365)	(0.0394)	(0.0451)	(0.0522)
Obs.	23482	20171	16816	13393	9787	6044	22370	19198	15983	12680	9181	5706
Mun.	3731	3377	2996	2592	2149	1656	3626	3274	2884	2507	2067	1617
Bwidth	IK	.85*IK	.70*IK	$.55^{*}IK$	.40*IK	.25*IK	IK	.85*IK	.70*IK	.55*IK	.40*IK	.25*IK

Panel A: Turnout and Blank Votes

Panel B: Spoilt Votes and Votes for Right-Wing Party

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	n spoilt	n spoilt	n spoilt	n spoilt	n spoilt	n spoilt	n right					
OL	$0.145^{**}$	$0.153^{***}$	$0.167^{***}$	$0.136^{**}$	0.110	-0.101	-0.0839	-0.0708	-0.162	-0.285	-0.223	-0.498
	(0.0573)	(0.0578)	(0.0603)	(0.0637)	(0.0703)	(0.0909)	(0.376)	(0.378)	(0.382)	(0.395)	(0.409)	(0.480)
Obs.	25484	21878	18178	14475	10609	6600	19279	16604	13782	10922	7948	4892
Mun.	3946	3580	3161	2721	2260	1734	3283	2978	2637	2300	1912	1514
Bwidth	IK	.85*IK	.70*IK	.55*IK	.40*IK	$.25^{*}IK$	IK	.85*IK	.70*IK	.55*IK	.40*IK	.25*IK

Panel C: Votes for Left- and Far-Left-Wing Parties

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	n left	n far left	n far left	n far left	n far left	n far left	n far left					
OL	0.395	0.366	0.300	0.482	0.406	0.482	0.0412	0.0101	0.0672	0.0721	0.127	0.0131
	(0.272)	(0.279)	(0.281)	(0.296)	(0.326)	(0.415)	(0.0841)	(0.0874)	(0.0928)	(0.102)	(0.111)	(0.129)
Obs.	16458	14086	11664	9146	6645	4150	17913	15285	12806	10091	7347	4547
Mun.	2957	2679	2393	2065	1742	1402	3367	3018	2706	2344	1940	1525
Bwidth	IK	.85*IK	.70*IK	.55*IK	.40*IK	.25*IK	IK	.85*IK	.70*IK	.55*IK	.40*IK	.25*IK

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants.

\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

#### 6.3 Pretrends in the Outcomes of Interest

The longitudinal structure of the data can be used to estimate dynamic causal effects. This subsection studies the timing of the effect of the electoral system on the outcomes of interest by estimating the following model:

$$y_{mt} = \alpha_m + \gamma_t + \beta_1 D_{mt} + \beta_2 D_{m,t+1} + f_1(x_{mt} - x^*) + f_2(x_{m,t+1} - x^*) + u_{mt}.$$
 (3)

If, conditional on the fixed effects, next period's electoral system is as good as random for those municipalities sufficiently close to the threshold, next period's electoral system should not help predict this period's outcome variable.<sup>34</sup> Thus, the coefficient  $\beta_2$  serves as a robustness check:  $\beta_2 \neq 0$  would suggest an endogeneity problem that may raise concerns about the validity of the approach of the paper. Alternatively, we can think of (3) as testing whether treatment and control municipalities were on different trends before the realization of this period's variables.<sup>35</sup>

Table 7 shows the results. The estimates for  $\beta_2$  are not statistically significant for *Turnout*. For some bandwidth choices, the coefficients for *Lists* are significant but they are small and have the opposite sign to the contemporaneous effect, suggesting that, if there is any difference, OL municipalities had *fewer* lists in the previous election. Moreover,  $\beta_1$ , the contemporaneous effect of the electoral system on the outcomes, remains similar to the one obtained in the baseline specifications, for the three outcome variables. That implies that previous trends in the outcome variables are not driving the results.

#### 6.4 Donut Regressions

This subsection considers donut regressions in the spirit of Barreca et al. (2011). The idea of this technique is to exclude observations very close to the thresholds, where sorting is more likely to occur. Given the nature of the sorting process of Spanish municipalities, it is likely that the sorting of municipalities is limited to a small window around the threshold: since the discontinuity in the density appears not because of outright manipulation of the population data, but as a consequence of registration decisions of citizens, it does not seem plausible that the population size of municipalities that are self-selecting into treatment is much bigger than what is strictly necessary to be above the threshold, or that municipalities with a population size far below the threshold attempt to cross it. Thus, finding a similar effect after municipalities close to the threshold are excluded would reinforce the credibility of the estimates.

For each outcome variable and bandwidth choice, I consider four specifications. The first one is the benchmark regression, in which no observations are excluded. The second excludes observations within a window of six inhabitants around the threshold (three at each side of the threshold), the third excludes those within an interval of 10 inhabitants and the fourth those within an interval of 20 inhabitants.

The results (see Table 8) show that excluding observations very close to the threshold does not affect the results, as they are remarkably similar in magnitude to the ones obtained in the baseline specifications. For voter turnout, the results are significant when up to 10 inhabitants around the threshold are excluded.<sup>36</sup> For a bandwidth of half that length, the results are significant even when municipalities with a population within a 20 inhabitants around the threshold are excluded. For the number of lists, the coefficients remain significant across all specifications. These results suggest that the findings of the paper are not being driven by municipalities of which concerns about self-selection could still persist.

<sup>&</sup>lt;sup>34</sup>According to Lee and Lemieux (2010), "finding a discontinuity in  $y_{mt}$  but not in  $y_{m,t-1}$  would be a strong piece of evidence supporting the validity of the RD design".

 $<sup>^{35}</sup>$ It is necessary to flexibly control for population in both periods (i.e. to include the functions  $f_1$  and  $f_2$ ) to capture the effect of the electoral system and not of changes in population size.

 $<sup>^{36}</sup>$ When 20 inhabitants are excluded the results are not significant but the point estimate remains similar: it is the increased standard errors due the decrease in sample size that makes the coefficient lose significance.

	(1)	(2)	(3)	(4)	(5)	(6)
	turnout	turnout	turnout	turnout	turnout	turnout
OL	$1.628^{***}$	$1.792^{***}$	$1.754^{***}$	$2.014^{***}$	$2.251^{***}$	$1.620^{**}$
	(0.410)	(0.425)	(0.443)	(0.471)	(0.548)	(0.762)
OL(+1)	-0.259	-0.302	-0.313	-0.191	-0.150	0.563
	(0.437)	(0.455)	(0.474)	(0.495)	(0.545)	(0.713)
Observations	14136	12005	9937	7792	5674	3516
Municipalities	2730	2507	2233	1974	1686	1405
Bandwidth	IK	.85*IK	$.70^{*}IK$	$.55^{*}IK$	$.40^{*}IK$	.25*IK

Panel	A:	Turnout
T OUTLOT		

Panel B: Lists

	(1)	(2)	(3)	(4)	(5)	(6)
	lists	lists	lists	lists	lists	lists
OL	$0.349^{***}$	$0.360^{***}$	$0.374^{***}$	$0.385^{***}$	$0.398^{***}$	$0.358^{***}$
	(0.0308)	(0.0319)	(0.0340)	(0.0366)	(0.0432)	(0.0576)
OL(+1)	-0.0458	$-0.0619^{**}$	$-0.0776^{**}$	$-0.0877^{***}$	$-0.104^{***}$	-0.0736
	(0.0281)	(0.0292)	(0.0308)	(0.0337)	(0.0362)	
Observations	15663	13426	11084	8723	6312	3962
Municipalities	2904	2649	2383	2087	1767	1466
Bandwidth	IK	.85*IK	.70*IK	.55*IK	$.40^{*}{ m IK}$	.25*IK

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants. \* p<0.10, \*\* p<0.05, \*\*\* p<0.01.

#### Table 8: Robustness: Donut Regressions

#### Panel A: Turnout

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	turnout	turnout	turnout	turnout	turnout	turnout	turnout	turnout	
OL	1.031***	$0.925^{**}$	0.920**	0.754	$1.676^{***}$	$1.546^{***}$	$1.689^{***}$	1.876***	
	(0.381)	(0.419)	(0.440)	(0.494)	(0.445)	(0.517)	(0.583)	(0.710)	
Observations	15954	15531	15249	14583	8039	7616	7334	6668	
Municipalities	2826	2825	2825	2825	2002	1994	1991	1976	
Bandwidth	IK	IK	IK	IK	$.50^{*}IK$	.50*IK	$.50^{*}IK$	$.50^{*}IK$	
Excluded	0	6	10	20	0	6	10	20	
Panel B: Lists									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	lists	lists	lists	lists	lists	lists	lists	lists	
OL	$0.314^{***}$	0.307***	$0.288^{***}$	0.282***	$0.361^{***}$	$0.357^{***}$	0.333***	0.353***	
	(0.0292)	(0.0312)	(0.0321)	(0.0351)	(0.0353)	(0.0389)	(0.0415)	(0.0486)	
Observations	17693	17270	16988	16322	9000	8577	8295	7629	
Municipalities	2993	2992	2992	2992	2102	2096	2093	2084	
Bandwidth	IK	IK	IK	IK	.50*IK	.50*IK	.50*IK	.50*IK	

20

0

6

10

20

Standard errors clustered by municipality in parentheses.

6

0

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants.

10

\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

Excluded

# 7 Conclusion

By exploiting the unique institutional framework of Spanish local elections, I have shown that the electoral system has an effect on voter turnout. In particular, an open list, plurality-at-large system increases turnout by between one and two percentage points with respect to a closed list proportional representation system. The results suggest that open list systems where parties run in candidate lists but voters can express their preferences for individual candidates are conducive to more voter turnout.

Working on understanding this type of electoral systems better could be a fruitful avenue for future research. Economists and political scientists have extensively worked on voter turnout and the motivations for the decision of voting or abstaining (Battaglini et al., 2010; Feddersen and Pesendorfer, 1996; Feddersen and Sandroni, 2006; Ledyard, 1984; Palfrey and Rosenthal, 1985), and this is still an active area of research (DellaVigna et al., 2013; Fujiwara et al., 2014; Gerber et al., 2008; Herrera et al., 2013; Kartal, 2013; Nickerson, 2008). However, to the best of my knowledge, we still do not have a theoretical framework to think about voter turnout in open list elections, in which there is competition both across and within parties.

# References

Ames, B. 1995a. "Electoral rules, constitutency pressures, and pork barrel: Bases of voting in the Brazilian congress. *The Journal of Politics*, 57 (4), 324-343.

Ames, B. 1995b. Electoral strategies under open-list proportional representation. *American Journal of Political Science*, 39 (2), 406-433.

Angrist, J. D. and Pischke, J. S. 2008. Mostly harmless econometrics: An empiricist's companion. *Princeton University Press.* 

Arnold, F., and Freier, R. 2013. Signature requirements and citi-Quasi-experimental evidence Germany. Available zen initatives: from at http://www.diw.de/documents/publikationen/73/diw\_01.c.424904.de /dp1311.pdf.

Barreca, A. I., Guldi, M., Lindo, J. M., and Waddell, G. R. 2011. Saving Babies? Revisiting the effect of very low birth weight classification. *The Quarterly Journal of Economics*, 126(4), 2117-2123.

Battaglini, Marco, Rebecca B. Morton, and Thomas R. Palfrey. 2010. The swing voter's curse in the laboratory. *The Review of Economic Studies*, 77(1), 61-89.

Blais, A. 2006. What affects voter turnout? Annual Review of Political Science, 9, 111-125.

Blais, A., and Aarts, K. 2006. Electoral systems and turnout. Acta Politica, 41(2), 180-196.

Blais, A., and Dobrzynska, A. 1998. Turnout in electoral democracies. *European Journal of Political Research*, 33(2), 239-261.

Bordignon, M., Nannicini, T., and Tabellini, G. 2013. Moderating political extremism: Single round vs runoff elections under plurality rule. *IZA Discussion Paper*.

Bowler, S., Donovan, T., and Brockington, D. 2003. Electoral reform and minority representation: Local experiments with alternative elections. *Ohio State University Press.* 

Brollo, F., Nannicini, T., Perotti, R., and Tabellini, G. 2013. The political resource curse. *The American Economic Review*, 103(5), 1759-1796.

Campa, P. 2012. Gender quotas, female politicians and public expenditures: quasi-experimental evidence. ECONPUBBLICA Working Paper No. 157.

Carey, J. M. and Shugart, M. S. 1995. Incentives to cultivate a personal vote: A rank ordering of electoral formulas. *Electoral Studies*, 14 (4), 417-439.

Casas-Arce, P., and Saiz, A. 2012. Women and power: Unwilling, ineffective, or held back? *IZA Discussion Paper*.

Chang, E. C. C. 2005. Electoral incentives for political corruption under open-list proportional representation. *The Journal of Politics*, 67 (3), 716-730.

Chang, E. C. C. and Golden, M. A. 2007. Electoral systems, district magnitude and corruption. *British Journal of Political Science*, 37 (1), 115-137.

DellaVigna, S., J. List, U. Malmendier, and G. Rao. 2013. "Voting to tell others." *Mimeo*, U.C. Berkeley.

Dos Santos, A. M. 2007. Do electoral rules matter? Electoral list models and their effects on party competition and institutional performance. Available at http://socialsciences.scielo.org/pdf/s\_dados/v3nse/scs\_a06.pdf.

Downs, A. 1957. An economic theory of democracy. New York: Harper.

Egger, P., and Koethenbuerger, M. 2010. Government spending and legislative organization: Quasi-experimental evidence from Germany. *American Economic Journal: Applied Economics*, 200-212.

Eggers, A. C. 2013. Proportionality and turnout: Evidence from French municipalities. Available at http://andy.egge.rs/papers/Eggers\_ProportionalityTurnoutFrance.pdf.

Farrell, D. M., and McAllister, I. 2006. Voter satisfaction and electoral systems: Does preferential voting in candidate-centred systems make a difference? *European Journal of Political Research*, 45(5), 723-749.

Feddersen, Timothy J., and Wolfgang Pesendorfer. 1996. The swing voter's curse. The American Economic Review, 86(3), 408-424.

Feddersen, T., and Sandroni, A. 2006. A theory of participation in elections. *The American Economic Review*, 96(4), 1271-1282.

politicians: Ferraz. С., and Finan, F. 2009.Motivating The imof monetary incentives on quality and performance. Available pacts at http://eml.berkeley.edu/~ffinan/Finan\_MPoliticians.pdf.

Fornos, C. A., Power, T. J., and Garand, J. C. 2004. Explaining voter turnout in Latin America, 1980 to 2000. *Comparative Political Studies*, 37(8), 909-940.

Fujiwara, T. 2011. A Regression Discontinuity Test of Strategic Voting and Duverger's Law. *Quarterly Journal of Political Science*, 6(3-4), 197-233.

Fujiwara, T. 2015. Voting Technology, Political Responsiveness, and Infant Health: Evidence From Brazil. *Econometrica*, 83: 423464.

Fujiwara, T., Meng, K. C., and Vogl, T. 2014. Estimating habit formation in voting. Available at http://www.princeton.edu/~fujiwara/papers/fmv\_habit\_formation\_voting\_feb2014.pdf.

Gagliarducci, S., and Nannicini, T. 2013. Do better paid politicians perform better? Disentangling incentives from selection. *Journal of the European Economic Association*, 11, 369-398.

Gerber, A.S., D.P. Green, and C.W. Larimer. 2008. "Social pressure and voter turnout: Evidence from a large-scale field experiment." *American Political Science Review*, 102(1), 33-48.

Geys, B. 2006. Explaining voter turnout: A review of aggregate-level research. *Electoral Studies*, 25(4), 637-663.

Grembi, V., Nannicini, T., and Troiano, U. 2012. Policy responses to fiscal restraints: A difference-in-discontinuities design. Available at http://www.igier.unibocconi.it/files/397.pdf.

Herrera, H., Morelli, M., and Palfrey, T. R. 2013. Turnout and power sharing. *The Economic Journal*, 124 (574), F131–F162.

Hinnerich, B. T. and P. Pettersson-Lidbom. 2014. Democracy, redistribution, and political participation: Evidence from Sweden 1919-1938. *Econometrica* 82 (3), 961{993.

Hirota, H., and Yunoue, H. 2013. Does local council size affect land development expenditure? Quasi-experimental evidence from Japanese municipal data. Available at http://mpra.ub.uni-muenchen.de/43723/.

Hix, S., and Hagemann, S. 2009. Could changing the electoral rules fix European parliament elections? *Politique européenne*, (2), 37-52.

Imbens, G., and Kalyanaraman, K. 2012. Optimal bandwidth choice for the regression discontinuity estimator. *The Review of Economic Studies*, 79(3), 933-959.

Imbens, G. W. and T. Lemieux (2008). Regression discontinuity designs: A guide to practice. *Journal of econometrics*, 142 (2), 615-635.

Jackman, R. W. 1987. Political institutions and voter turnout in the industrial democracies. *The American Political Science Review*, 81(2), 405-423.

Jackman, R. W., and Miller, R. A. 1995. Voter turnout in the industrial democracies during the 1980s. *Comparative Political Studies*, 27(4), 467-492.

Kartal, M. 2013. Welfare comparison of proportional representation and majoritarian rule with endogenous turnout. Available at https://files.nyu.edu/mk2672/public/.

Karvonen, L. 2004. Preferential voting: incidence and effects. *International Political Science Review*, 25(2), 203-226.

Ladner, A., and Milner, H. 1999. Do voters turn out more under proportional than majoritarian systems? The evidence from Swiss communal elections. *Electoral Studies*, 18(2), 235-250.

Ledyard, J. O. 1984. The pure theory of large two-candidate elections. *Public Choice*, 44(1), 7-41.

Lee, D. S., and Lemieux, T. 2010. Regression discontinuity designs in economics. *The Journal of Economic Literature*, 48(2), 281-355.

Lemieux, T., and Milligan, K. 2008. Incentive effects of social assistance: A regression discontinuity approach. *Journal of Econometrics*, 142(2), 807-828.

Litschig, S., and Morrison, K. 2013. The Impact of Intergovernmental Transfers on Education Outcomes and Poverty Reduction. *AEJ: Applied Economics*, 5(4): 206-240.

Mattila, M. 2003. Why bother? Determinants of turnout in the European elections. *Electoral Studies*, 22(3), 449-468.

McCrary, J. 2008. Manipulation of the running variable in the regression discontinuity design: A density test. *Journal of Econometrics*, 142(2), 698-714.

Milner, H., and Ladner, A. 2006. Can PR voting serve as a shelter against declining turnout? Evidence from Swiss municipal elections. *International Political Science Review*, 27(1), 29-45.

Negri, M. 2014. Minority representation in proportional representation systems. Available at http://margheritanegri.files.wordpress.com/2013/03/minority-representation-in-pr-systems-negri1.pdf.

Nickerson, D.W. 2008. "Is voting contagious? Evidence from two field experiments." *American Political Science Review*, 102(1), 49-57.

Palfrey, Thomas R., and Howard Rosenthal. 1985. Voter participation and strategic uncertainty. *The American Political Science Review* (1985): 62-78.

Persson, T., Tabellini, G., and Trebbi, F. 2003. Electoral rules and corruption. *Journal of the European Economic Association*, 1(4), 958-989.

Pettersson-Lidbom, P. 2012. Does the size of the legislature affect the size of government: Evidence from two natural experiments. *Journal of Public Economics*, Volume 96, Issues 3–4, 269-278.

Riker, W. H., and Ordeshook, P. C. 1968. A theory of the calculus of voting. *The American Political Science Review*, 25-42.

Robbins, J. W. 2010. The personal vote and voter turnout. *Electoral Studies*, 29(4), 661-672.

Sanz, C. 2014. Does direct democracy reduce deficits? Evidence from Spain. Princeton University Working Paper.

Schram, A., and Sonnemans, J. 1996. Voter turnout as a participation game: An experimental investigation. *International Journal of Game Theory*, 25(3), 385-406.

Shugart, M. S., M. E. Valdini, and Suominen, K. 2005. Looking for locals: Voter information demands and personal vote-earning attributes of legislators under proportional representation. *American Journal of Political Science*, 49 (2), 437-449.

Selb, P. 2009. A deeper look at the proportionality-turnout nexus. *Comparative Political Studies*, 42(4), 527-548.

St-Vincent, S. L. 2013. An experimental test of the pivotal voter model under plurality and PR elections. *Electoral Studies*, 32(4), 795–806.

Sweeting, D. 2009. The institutions of strong local political leadership in Spain. *Environment and planning C: government and policy 27.* 

# A Robustness Checks for the 100-inhabitant Threshold

In this appendix I show robustness checks for the results at the 100-inhabitant threshold. Table 9 tests for pretrends and Table 10 shows the results of the donut regression discontinuity design.

	(1)	(2)	(3)	(4)	(5)	(6)
	turnout	turnout	turnout	turnout	turnout	turnout
OL	$3.177^{***}$	$3.195^{***}$	$3.082^{***}$	$2.974^{***}$	$2.796^{***}$	$2.159^{**}$
	(0.588)	(0.600)	(0.621)	(0.672)	(0.764)	(0.978)
OL(+1)	0.152	0.300	0.235	0.285	0.450	0.192
	(0.544)	(0.563)	(0.577)	(0.626)	(0.661)	(0.774)
Observations	8742	7555	6263	4915	3588	2242
Municipalities	2504	2220	1921	1588	1256	931
Bandwidth	IK	85%	70%	55%	40%	25%

Panel A: Turnout

Panel B: Lists

	(1)	(2)	(3)	(4)	(5)	(6)
	lists	lists	lists	lists	lists	lists
OL	0.100**	0.0927**	$0.0827^{*}$	0.0538	0.0679	0.223**
	(0.0439)	(0.0463)	(0.0496)	(0.0555)	(0.0657)	(0.0981)
OL(+1)	-0.0554	-0.0452	-0.0592	-0.0653	-0.000580	0.0453
	(0.0434)	(0.0468)	(0.0486)	(0.0533)	(0.0583)	(0.0718)
Observations	5401	4596	3813	2999	2146	1341
Municipalities	1694	1513	1316	1117	911	677
Bandwidth	IK	85%	70%	55%	40%	25%

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants. \* p<0.10, \*\* p<0.05, \*\*\* p<0.01.

Panel A: Turnout

	(1)	(2)	(2)		(=)	( )		(2)	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	turnout	turnout	turnout	turnout	turnout	turnout	turnout	turnout	
OL	$2.811^{***}$	2.947***	2.983***	$3.448^{***}$	2.487***	$2.555^{***}$	2.484***	$3.679^{***}$	
	(0.521)	(0.578)	(0.639)	(0.803)	(0.595)	(0.696)	(0.804)	(1.074)	
Obs.	11161	10739	10493	9871	5737	5315	5069	4447	
Mun.	2621	2621	2621	2608	1603	1603	1603	1589	
Bw.	IK	IK	IK	IK	50%	50%	50%	50%	
Excluded	0	6	10	20	0	6	10	20	
Panel B: Party-Lists									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	

	(1)	(2)	$(\mathbf{U})$	(4)	( <b>0</b> )	(0)	(1)	(0)
	lists	lists	lists	lists	lists	lists	lists	lists
OL	$0.0778^{**}$	$0.0818^{*}$	0.101**	$0.147^{**}$	0.0517	0.0506	0.0941	0.187
	(0.0390)	(0.0475)	(0.0482)	(0.0669)	(0.0506)	(0.0730)	(0.0827)	(0.145)
Obs.	6907	6485	6239	5617	3571	3149	2903	2281
Mun.	1823	1823	1823	1810	1190	1184	1180	1144
Bw.	IK	IK	IK	IK	50%	50%	50%	50%
Excluded	0	6	10	20	0	6	10	20

Standard errors clustered by municipality in parentheses.

All regressions include municipality and year fixed effects.

IK refers to the minimum of Imbens and Kalyanaraman's (2012) bandwidth and 250 inhabitants.

\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01.

#### **BANCO DE ESPAÑA PUBLICATIONS**

#### WORKING PAPERS

- 1511 PATRICIA GÓMEZ-GONZÁLEZ: Financial innovation in sovereign borrowing and public provision of liquidity.
- 1512 MIGUEL GARCÍA-POSADA and MARCOS MARCHETTI: The bank lending channel of unconventional monetary policy: the impact of the VLTROs on credit supply in Spain.
- 1513 JUAN DE LUCIO, RAÚL MÍNGUEZ, ASIER MINONDO and FRANCISCO REQUENA: Networks and the dynamics of firms' export portfolio.
- 1514 ALFREDO IBÁÑEZ: Default near-the-default-point: the value of and the distance to default.
- 1515 IVÁN KATARYNIUK and JAVIER VALLÉS: Fiscal consolidation after the Great Recession: the role of composition.
- 1516 PABLO HERNÁNDEZ DE COS and ENRIQUE MORAL-BENITO: On the predictability of narrative fiscal adjustments.
- 1517 GALO NUÑO and CARLOS THOMAS: Monetary policy and sovereign debt vulnerability.
- 1518 CRISTIANA BELU MANESCU and GALO NUÑO: Quantitative effects of the shale oil revolution.
- 1519 YAEL V. HOCHBERG, CARLOS J. SERRANO and ROSEMARIE H. ZIEDONIS: Patent collateral, investor commitment and the market for venture lending.
- 1520 TRINO-MANUEL ÑÍGUEZ, IVAN PAYA, DAVID PEEL and JAVIER PEROTE: Higher-order risk preferences, constant relative risk aversion and the optimal portfolio allocation.
- 1521 LILIANA ROJAS-SUÁREZ and JOSÉ MARÍA SERENA: Changes in funding patterns by Latin American banking systems: how large? how risky?
- 1522 JUAN F. JIMENO: Long-lasting consequences of the European crisis.
- 1523 MAXIMO CAMACHO, DANILO LEIVA-LEON and GABRIEL PEREZ-QUIROS: Country shocks, monetary policy expectations and ECB decisions. A dynamic non-linear approach.
- 1524 JOSÉ MARÍA SERENA GARRALDA and GARIMA VASISHTHA: What drives bank-intermediated trade finance? Evidence from cross-country analysis.
- 1525 GABRIELE FIORENTINI, ALESSANDRO GALESI and ENRIQUE SENTANA: Fast ML estimation of dynamic bifactor models: an application to European inflation.
- 1526 YUNUS AKSOY and HENRIQUE S. BASSO: Securitization and asset prices.
- 1527 MARÍA DOLORES GADEA, ANA GÓMEZ-LOSCOS and GABRIEL PEREZ-QUIROS: The Great Moderation in historical perspective. Is it that great?
- 1528 YUNUS AKSOY, HENRIQUE S. BASSO, RON P. SMITH and TOBIAS GRASL: Demographic structure and macroeconomic trends.
- 1529 JOSÉ MARÍA CASADO, CRISTINA FERNÁNDEZ and JUAN F. JIMENO: Worker flows in the European Union during the Great Recession.
- 1530 CRISTINA FERNÁNDEZ and PILAR GARCÍA PEREA: The impact of the euro on euro area GDP per capita.
- 1531 IRMA ALONSO ÁLVAREZ: Institutional drivers of capital flows.
- 1532 PAUL EHLING, MICHAEL GALLMEYER, CHRISTIAN HEYERDAHL-LARSEN and PHILIPP ILLEDITSCH: Disagreement about inflation and the yield curve.
- 1533 GALO NUÑO and BENJAMIN MOLL: Controlling a distribution of heterogeneous agents.
- 1534 TITO BOERI and JUAN F. JIMENO: The unbearable divergence of unemployment in Europe.
- 1535 OLYMPIA BOVER: Measuring expectations from household surveys: new results on subjective probabilities of future house prices.
- 1536 CRISTINA FERNÁNDEZ, AITOR LACUESTA, JOSÉ MANUEL MONTERO and ALBERTO URTASUN: Heterogeneity of markups at the firm level and changes during the great recession: the case of Spain.
- 1537 MIGUEL SARMIENTO and JORGE E. GALÁN: The influence of risk-taking on bank efficiency: evidence from Colombia.
- 1538 ISABEL ARGIMÓN, MICHEL DIETSCH and ÁNGEL ESTRADA: Prudential filters, portfolio composition and capital ratios in European banks.
- 1539 MARIA M. CAMPOS, DOMENICO DEPALO, EVANGELIA PAPAPETROU, JAVIER J. PÉREZ and ROBERTO RAMOS: Understanding the public sector pay gap.
- 1540 ÓSCAR ARCE, SAMUEL HURTADO and CARLOS THOMAS: Policy spillovers and synergies in a monetary union.
- 1601 CHRISTIAN CASTRO, ÁNGEL ESTRADA and JORGE MARTÍNEZ: The countercyclical capital buffer in Spain: an analysis of key guiding indicators.

- 1602 TRINO-MANUEL ÑÍGUEZ and JAVIER PEROTE: Multivariate moments expansion density: application of the dynamic equicorrelation model.
- 1603 ALBERTO FUERTES and JOSÉ MARÍA SERENA: How firms borrow in international bond markets: securities regulation and market segmentation.
- 1604 ENRIQUE ALBEROLA, IVÁN KATARYNIUK, ÁNGEL MELGUIZO and RENÉ OROZCO: Fiscal policy and the cycle in Latin America: the role of financing conditions and fiscal rules.
- 1605 ANA LAMO, ENRIQUE MORAL-BENITO and JAVIER J. PÉREZ: Does slack influence public and private labour market interactions?
- 1606 FRUCTUOSO BORRALLO, IGNACIO HERNANDO and JAVIER VALLÉS: The effects of US unconventional monetary policies in Latin America.
- 1607 VINCENZO MERELLA and DANIEL SANTABÁRBARA: Do the rich (really) consume higher-quality goods? Evidence from international trade data.
- 1608 CARMEN BROTO and MATÍAS LAMAS: Measuring market liquidity in US fixed income markets: a new synthetic indicator.
- 1609 MANUEL GARCÍA-SANTANA, ENRIQUE MORAL-BENITO, JOSEP PIJOAN-MAS and ROBERTO RAMOS: Growing like Spain: 1995-2007.
- 1610 MIGUEL GARCÍA-POSADA and RAQUEL VEGAS: Las reformas de la Ley Concursal durante la Gran Recesión.
- 1611 LUNA AZAHARA ROMO GONZÁLEZ: The drivers of European banks' US dollar debt issuance: opportunistic funding in times of crisis?
- 1612 CELESTINO GIRÓN, MARTA MORANO, ENRIQUE M. QUILIS, DANIEL SANTABÁRBARA and CARLOS TORREGROSA: Modelling interest payments for macroeconomic assessment.
- 1613 ENRIQUE MORAL-BENITO: Growing by learning: firm-level evidence on the size-productivity nexus.
- 1614 JAIME MARTÍNEZ-MARTÍN: Breaking down world trade elasticities: a panel ECM approach.
- 1615 ALESSANDRO GALESI and OMAR RACHEDI: Structural transformation, services deepening, and the transmission of monetary policy.
- 1616 BING XU, ADRIAN VAN RIXTEL and HONGLIN WANG: Do banks extract informational rents through collateral?
- 1617 MIHÁLY TAMÁS BORSI: Credit contractions and unemployment.
- 1618 MIHÁLY TAMÁS BORSI: Fiscal multipliers across the credit cycle.
- 1619 GABRIELE FIORENTINI, ALESSANDRO GALESI and ENRIQUE SENTANA: A spectral EM algorithm for dynamic factor models.
- 1620 FRANCISCO MARTÍ and JAVIER J. PÉREZ: Spanish public finances through the financial crisis.
- 1621 ADRIAN VAN RIXTEL, LUNA ROMO GONZÁLEZ and JING YANG: The determinants of long-term debt issuance by European banks: evidence of two crises.
- 1622 JAVIER ANDRÉS, ÓSCAR ARCE and CARLOS THOMAS: When fiscal consolidation meets private deleveraging.
- 1623 CARLOS SANZ: The effect of electoral systems on voter turnout: evidence from a natural experiment.

# BANCODE ESPAÑA

Eurosistema

Unidad de Servicios Auxiliares Alcalá, 48 - 28014 Madrid E-mail: publicaciones@bde.es www.bde.es